Author’s response to reviews

Title: Effectiveness and safety assessment of drospirenone/ethinyl estradiol tablet in treatment of PCOS patients: a single center, prospective, observational study

Authors:

Li Li (lili1406@yahoo.com)
Ruiqin Zhang (355406868@qq.com)
Jing Zeng (zj13829734741@163.com)
Hu Ke (13662328522@126.com)
Xiuhong Peng (pengxiuh@163.com)
Liying Huang (1139495703@qq.com)
Hongmei Zhang (1632205725@qq.com)
Zhijing Chen (chenzhijing136@163.com)
Tian Li (ltt305@qq.com)
Qiuxiao Tan (751156734@qq.com)
Ying Yang (452913560@qq.com)
Xiaofang Li (1251939704@qq.com)
Xin Li (lxsure@fudan.edu.cn)

Version: 3 Date: 11 Feb 2020

Author’s response to reviews:

1. We note that XI is include twice in the Authors’ Contributions section. As such, please distinguish between them by designating them with numbers 1 and 2. For example: YZ1 and YZ2. YZ1 would correspond to the author furthest up on the author list.

We also note that HZ is missing from this section, but is included on the title page. Therefore, please ensure that all authors are included in this section.
Response: We thank the editor for this comment. All the authors have been duly added in the author contribution section (page#14).

2. As a reminder, please proofread and ensure that when you upload your revised submission it is in the final form for publication. Please remove any tracked changes or highlighting and include only a single
clean copy of the manuscript. Should you wish to respond to these revision requests, please include the information in the designated input box only.

Response: We thank the editor for this comment. The manuscript has been submitted as a clean copy and there are no highlighted sections or track changes. The point-wise response has been provided in the designated box during submission.